Author's response to reviews

Title: Screening for Iron Deficiency and Iron Deficiency Anaemia in Pregnancy: A Structured Review and Gap Analysis Against UK National Screening Criteria

Authors:

Ruramayi A Rukuni (ruramayirukuni@gmail.com)
Marian Knight (Marian.Knight@npeu.ox.ac.uk)
Michael M Murphy (Mike.Murphy@nhsbt.nhs.uk)
David Roberts (David.Roberts@ndcls.ox.ac.uk)
Simon Stanworth (Simon.Stanworth@nhsbt.nhs.uk)

Version: 2 Date: 20 March 2015

Author's response to reviews:

The following revisions have been made:

1) Line and page numbers have been added to the manuscript to facilitate peer review.

2) The email addresses of each co-author have been added to the title page.

3) An ethics statement has been included after the Methods section of the manuscript.